Author's response to reviews

Title: Mental Health, Sexual Identity, and Interpersonal Violence: Findings from the Australian Longitudinal Women's Health Study

Authors:

Laura Szalacha (drlszalacha@email.arizona.edu)

Tonda Hughes (thughes@uic.edu)

Ruth McNair (r.mcnair@unimelb.edu.au)

Deborah Loxton (deborah.loxton@newcastle.edu.au)

Version: 2 Date: 31 Aug 2017

Author’s response to reviews:

Dear Mr. Sisperez,

Thank you for the opportunity to make the necessary revisions to our manuscript. I have delineated below each change made.

1. Please include the email addresses of all authors on the title page of your manuscript.

All four email addresses are on the title page of the manuscript.

2. Please include a "Declarations" section at the end of your manuscript. Contained within this section should be a list of abbreviations, as well as sections on 'Ethics approval and consent to participate', 'consent for publication', 'Availability of data and materials', Competing interests', 'Funding', 'Authors' Contributions' and 'Acknowledgements'. In the event that any of these sections are not relevant for your study, we would still ask you to include them, but to include a statement detailing that this information is not relevant. Further details on the information that should be included in your "Declarations" section, as well as how to format it, can be found in our submission guidelines:

https://bmcwomenshealth.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Please pay particular attention to the Ethics section and ensure to include the full, official name of the ethics committee that approved your study.

We included a Declarations section at the end of the manuscript. Contained within this section is a list of acknowledgements, funding, abbreviations, ethics approval and consent to participate,
consent for publication, availability of data and materials, competing interests and authors' contributions.

The Ethics section includes the full, official name of the ethics committees that approved the study and their numbers.

3. Please add a "Background" section heading to your main manuscript.

We have added a "Background" section heading to the manuscript.

I hope that everything is in proper order.

Thank you very much,

Dr. Laura Szalacha